Author’s response to reviews

Title: The performance of rapid plasma reagin (RPR) titer in HIV-negative general paresis after neurosyphilis therapy

Authors:

Ying Jiang (jiangying722@163.com)
Ruihui Weng (13825095630@126.com)
Yuefeng Zhang (zhangyuefengbh@163.com)
Rong Fan (517199864@qq.com)
Yulun Liu (yulunliu@mail.med.upenn.edu)
Zhigang Chen (983822418@qq.com)
Fuhua Peng (pengfuhuazssy@163.com)
Yong Chen (ychen123@mail.med.upenn.edu)
Xiaohong Chen (xiaohongchenzssy@aliyun.com)

Version: 3 Date: 19 Feb 2018

Author’s response to reviews:

Dear editor and reviewers: Firstly, thank you for your comments. According to your comments, I want to do some replies. And we have revised the changes in the manuscript by highlighting with red words. Reviewer reports: Hanna Jarva (Reviewer 2): This revised version is much better than the previous ones. However, there are still grammatical mistakes. I suggest that the authors have the language checked by a native English speaker. Answer: Many thanks for your comment. We have carefully corrected the grammatical errors and polished the writing again. It should be more readable and more scientific. We sincerely hope that the revision is satisfactory. Best regards Xiaohong Chen and Ying Jiang Department of Neurology The Third Affiliated Hospital, Sun Yat-sen University, 600 Tianhe Road, Guangzhou, Guangdong 510630 China. Tel/Fax: +86-20-85513761 Email: xiaohongchenzssy@aliyun.com jiangying722@163.com;